Author's response to reviews

Title: A modified surgical technique in the management of eyelid burns: a case series

Haiying Liu 1, MD, Qigang Wang 1, MD, Shudong Sun 1, MD, Youxin Ji 2*, MD

1 Burn Center, Weifang People’s Hospital, Weifang Medical School, Weifang, Shandong Province 265000, PR China
2 Department of Surgery, Qingdao Central Hospital, Qingdao, Shandong Province 266042, PR China

*Corresponding author YJ: kevinji78@gmail.com HL: hengtrade@yahoo.com QW: mdxue001@gmail.com SD: 123456789ji@gmail.com

Authors:

haiying liu (hengtrade@yahoo.com)
qigang wang (mdxue001@gmail.com)
shudong sun (123456789ji@gmail.com)
youxin ji (kevinji78@gmail.com)

Version: 6 Date: 11 May 2011

Author’s response to reviews:

patient's consent copy and corp pictures are included.